258' Copyright Ron Burback 1997JunctionsRon BurbackLecture 13

259' Copyright Ron Burback 1997JunctionsnThe CDS is not just one name space but a collection of name spaces spliced together at junction points. n/.:/fs - managed by DFSn/.:/sec - managed by securityn/.:/hosts/host_name/config managed by dced

260' Copyright Ron Burback 1997Why Junctions?nThe power of a unified name space. example: URLs (almost)nIf you can get to the name, you should be able to get to the thing. nI can ACL set on CDS, Security, and DFS objects with the same commands.

261' Copyright Ron Burback 1997It would be nice if ....nWe had a database junction.nemail folder junctionnURLs done right